Author’s response to reviews

Title: The potential impact of food taxes and subsidies on cardiovascular disease and diabetes burden and disparities in the United States

Authors:

Jose Penalvo (jose.penalvo@tufts.edu)
Frederick Cudhea (Frederick.Cudhea@tufts.edu)
Renata Micha (renata.micha@tufts.edu)
Colin Rehm (colin.rehm@gmail.com)
Ashkan Afshin (aafshin@uw.edu)
Parke Wilde (Parke.Wilde@tufts.edu)
Martin O'Flaherty (M.OFlaherty@liverpool.ac.uk)
Jonathan Pearson-Stuttard (j.pearson-stuttard@imperial.ac.uk)
Laurie Whitsel (Laurie.Whitsel@heart.org)
Thomas Gaziano (TGAZIANO@PARTNERS.ORG)
Simon Capewell (capewell@liverpool.ac.uk)
Dariush Mozaffarian (dariush.mozaffarian@tufts.edu)

Version: 1 Date: 10 Apr 2017

Author’s response to reviews:

Please find attached a document with the response to the reviewers